Author's response to reviews

Title: Phono-spectrographic analysis of heart murmurs in children

Authors:

Anna-Leena Noponen (anna-leena.noponen@pp.inet.fi)
Sakari Lukkarinen (sakari.lukkarinen@comsol.fi)
Anna Angerla (anu.angerla@hotmail.com)
Raimo Sepponen (raimo.sepponen@hut.fi)

Version: 4 Date: 7 May 2007

Author's response to reviews:

Dear Editor,

the following additions to the manuscript have been made:

1. A statement of the approval of an appropriate ethics committee is added to the Methods - section.

2. Competing interests section (none to declare) is added after Conclusions.

3. The manuscript formatting is reviewed according to Authors' checklist for manuscript.

We hope that everything is now ready for publishing and are waiting your instructions for the payment.

Sincerely,

Anna-Leena Noponen MD